\begin{thebibliography}{7}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\expandafter\ifx\csname bibnamefont\endcsname\relax
  \def\bibnamefont#1{#1}\fi
\expandafter\ifx\csname bibfnamefont\endcsname\relax
  \def\bibfnamefont#1{#1}\fi
\expandafter\ifx\csname citenamefont\endcsname\relax
  \def\citenamefont#1{#1}\fi
\expandafter\ifx\csname url\endcsname\relax
  \def\url#1{\texttt{#1}}\fi
\expandafter\ifx\csname urlprefix\endcsname\relax\def\urlprefix{URL }\fi
\providecommand{\bibinfo}[2]{#2}
\providecommand{\eprint}[2][]{\url{#2}}

\bibitem[{\citenamefont{Bloembergen}(1961)}]{bloembergen1961}
\bibinfo{author}{\bibfnamefont{N.}~\bibnamefont{Bloembergen}},
  \emph{\bibinfo{title}{Nuclear Magnetic Relaxation}} (\bibinfo{publisher}{W.A.
  Benjamine}, \bibinfo{year}{1961}).

\bibitem[{\citenamefont{Lee and Anderson}(1967)}]{crc_nmr}
\bibinfo{author}{\bibfnamefont{K.}~\bibnamefont{Lee}} \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{W.}~\bibnamefont{Anderson}},
  \emph{\bibinfo{title}{Nuclear spins, moments, and magnetic resonance
  frequencies}} (\bibinfo{publisher}{CRC Press}, \bibinfo{year}{1967}).

\bibitem[{\citenamefont{Cohen-Tannoudji
  et~al.}(2006)\citenamefont{Cohen-Tannoudji, Diu, and Laloe}}]{tannoudji2006}
\bibinfo{author}{\bibfnamefont{C.}~\bibnamefont{Cohen-Tannoudji}},
  \bibinfo{author}{\bibfnamefont{B.}~\bibnamefont{Diu}}, \bibnamefont{and}
  \bibinfo{author}{\bibfnamefont{F.}~\bibnamefont{Laloe}},
  \emph{\bibinfo{title}{Quantum Mechanics}}
  (\bibinfo{publisher}{Wiley-Interscience}, \bibinfo{year}{2006}).

\bibitem[{\citenamefont{lab staff}(2008)}]{nmrlg}
\bibinfo{author}{\bibfnamefont{J.}~\bibnamefont{lab staff}},
  \emph{\bibinfo{title}{Pulsed NMR lab guide}} (\bibinfo{year}{2008}).

\bibitem[{\citenamefont{Hahn}(1950)}]{hahn1950}
\bibinfo{author}{\bibfnamefont{E.}~\bibnamefont{Hahn}}, \bibinfo{journal}{Phys.
  Rev.}  (\bibinfo{year}{1950}).

\bibitem[{\citenamefont{Sewell}(2005)}]{nmrlg_old}
\bibinfo{author}{\bibfnamefont{S.}~\bibnamefont{Sewell}},
  \emph{\bibinfo{title}{Pulsed Nuclear Magnetic Resonance}}
  (\bibinfo{publisher}{MIT Copy Tech}, \bibinfo{year}{2005}).

\bibitem[{wat()}]{water-glycerine}
\emph{\bibinfo{title}{Viscosity of aqueous glycerine solutions}},
  \bibinfo{howpublished}{\url{http://www.dow.com/glycerine/resources/table18.h%
tm}}.

\end{thebibliography}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Sample template for MIT Junior Lab Student Written Summaries
% Available from http://web.mit.edu/8.13/www/Samplepaper/sample-paper.tex
%
% Last Updated April 12, 2007
%
% Adapted from the American Physical Societies REVTeK-4 Pages
% at http://publish.aps.org
%
% ADVICE TO STUDENTS: Each time you write a paper, start with this
%    template and save under a new filename.  If convenient, don't
%    erase unneeded lines, just comment them out.  Often, they
%    will be useful containers for information.
%
% Using pdflatex, images must be either PNG, GIF, JPEG or PDF.
%     Turn eps to pdf using epstopdf.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% PREAMBLE
% The preamble of a LaTeX document is the set of commands that precede
% the \begin{document} line.  It contains a \documentclass line
% to load the REVTeK-4 macro definitions and various \usepackage
% lines to load other macro packages.
%
% ADVICE TO STUDENTS: This preamble contains a suggested set of
%     class options to generate a ``Junior Lab'' look and feel that
%     facilitate quick review and feedback from one's peers, TA's
%     and section instructors.  Don't make substantial changes without
%     first consulting your section instructor.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\documentclass[aps,twocolumn,secnumarabic,balancelastpage,amsmath,amssymb,nofootinbib]{revtex4}

% Documentclass Options
    % aps, prl stand for American Physical Society and Physical Review Letters respectively
    % twocolumn permits two columns, of course
    % nobalancelastpage doesn't attempt to equalize the lengths of the two columns on the last page
        % as might be desired in a journal where articles follow one another closely
    % amsmath and amssymb are necessary for the subequations environment among others
    % secnumarabic identifies sections by number to aid electronic review and commentary.
    % nofootinbib forces footnotes to occur on the page where they are first referenced
        % and not in the bibliography
    % REVTeX 4 is a set of macro packages designed to be used with LaTeX 2e.
        % REVTeX is well-suited for preparing manuscripts for submission to APS journals.


\usepackage{lgrind}        % convert program listings to a form includable in a LaTeX document
\usepackage{chapterbib}    % allows a bibliography for each chapter (each labguide has it's own)
\usepackage{color}         % produces boxes or entire pages with colored backgrounds
\usepackage{graphics}      % standard graphics specifications
\usepackage[pdftex]{graphicx}      % alternative graphics specifications
\usepackage{longtable}     % helps with long table options
\usepackage{epsf}          % old package handles encapsulated post script issues
\usepackage{bm}            % special 'bold-math' package
%\usepackage{asymptote}     % For typesetting of mathematical illustrations
\usepackage{thumbpdf}
\usepackage[colorlinks=true]{hyperref}  % this package should be added after all others
                                        % use as follows: \url{http://web.mit.edu/8.13}
\usepackage{multirow}

%
% And now, begin the document...
% Students should not have to alter anything above this line
%

\newcommand{\bra}[1]{\left\langle #1 \right|}
\newcommand{\ket}[1]{\left|#1\right\rangle}
\newcommand{\braket}[2]{\left\langle#1 |  #2\right\rangle}

\begin{document}
\title{Pulsed NMR: Relaxation times as function of viscocity and impurities}
\author{Tony Hyun Kim}
%\email{kimt@mit.edu}
%\homepage{http://web.mit.edu/kimt/www/}
\date{\today}
\affiliation{MIT Department of Physics}


\begin{abstract}
We measure the spin-lattice ($T_1$) and spin-spin ($T_2$) relaxation times of a sample of nuclear spins, as a function of viscosity and impurity concentration. In the viscosity study, we confirm Bloembergen's observation of an inverse-law relationship between relaxation times and viscosity. The impurity investigation remains inconclusive. In general, our measurements support the claim that microscopic complexity (increased viscosity; additional impurities) increases relaxation rates.
\end{abstract}

\maketitle

\section{Introduction}

Modern attempts to utilize quantum mechanics for novel applications (e.g. quantum computing) are severely limited by the \emph{relaxation times} associated with the particular implementation. Relaxation times describe the timescale for which the system remains under coherent control by the experimenter. In this paper, we investigate the dynamics of an ensemble of nuclear spins as a prototype for the relaxation phenomenon.

We manipulate a sample of nuclear spins by the techniques of nuclear magnetic resonance (NMR) developed by Felix Bloch and E.M. Purcell, through which the spin ensemble can be perturbed away from thermal equilibrium. Following the excitation, we observe an exponential decay in the fraction of the sample following coherent evolution, which is the relaxation effect. We have measured relaxation along two dimensions: the \emph{spin-lattice} relaxation time ($T_1$) which describes the return to thermal occupation of the eigenstates; and the \emph{spin-spin} relaxation time ($T_2$) which is the disappearance of the coherent evolution due to ``dephasing'' among the individual spins. In an upcoming section, we will explain more explicitly the two effects in the context of a spin system.

We have considered the relaxation times $T_1$, $T_2$ as a function of viscosity of the sample, and as a function of magnetic impurity concentration. In both cases, we observe that relaxation times are shortened (more rapid decay) when additional interactions are added (i.e. more viscous; additional impurities). In the viscosity study, we have verified Bloembergen's observation of an inverse-law relationship.\cite{bloembergen1961}

\section{Theoretical background}

\subsection{Coherent manipulation of nuclear spins}

NMR experiments are conducted in a region of large bias field $B_0$. In our experiment, the nuclear spin is that of the proton $^1H$ in glycerine and water molecules. Because $^1H$ has total nuclear spin $I = 1/2$\cite{crc_nmr}, there are two possibilities for the spin projection: $\ket{+}$ (aligned with $B_0$) and $\ket{-}$ (anti-aligned). In thermal equilibrium, the sample occupies the eigenstates $\ket{+}$ and $\ket{-}$ in accordance with the Boltzmann distribution. 

The system is then perturbed by a small oscillating magnetic field $B_1 = B_{1,0}\cos(\omega t)$ in a direction perpendicular to the bias field $B_0$. The interaction between the applied field $\vec{B} = \vec{B_0} + \vec{B_1}$ and the nuclear spin is due to the magnetic Hamiltonian $H = -\vec{\mu} \cdot \vec{B}$, where $\vec{\mu} = \gamma \vec{I}$ is the magnetic moment associated with the spin, and $\gamma$ is the gyromagnetic ratio. In a static $B$-field, the (expectation of the) magnetic moment undergoes precessive motion about $B$ at the Larmor frequency $w_L = \gamma B_0$. The experimental signature of such collective precession is RF radiation, which is the basic detection mechanism in our experiment. The observed oscillatory response is called the \emph{free induction decay} (FID). The dynamics is more complicated for a time-dependent $B$. We give only a cursory analysis for the time-dependent case; a detailed and geometrically-enlightening discussion can be found in \cite{tannoudji2006,nmrlg}.

For our time-dependent field $\vec{B}(t) = B_1(t)\vec{e}_x + B_0\vec{e}_z$, we represent the spinor $\ket{\psi}$ in the basis of the static field eigenstates $\left\{\ket{+},\ket{-}\right\}$ as in:
\begin{equation}
	\ket{\psi(t)} = \left[\begin{array}{c}
												C_u(t)e^{+i\frac{\omega_L}{2}t}\\
												C_d(t)e^{-i\frac{\omega_L}{2}t}
												\end{array}\right]
	\label{eqn:rf_vector}
\end{equation}
where the exponential factor corresponds to the standard time-evolution due to the bias field $B_0$. (The splitting between $\ket{+}$ and $\ket{-}$ corresponds to the Larmor frequency.) It can be shown\cite{nmrlg} that the Schr\"{o}dinger equation in the above basis yields:
\begin{eqnarray}
	\dot{C}_u &=& \frac{i\omega_x}{2}C_d\left[e^{i(\omega-\omega_L)t}+e^{-i(\omega+\omega_L)t}\right]\\
	\dot{C}_d &=& -\frac{i\omega_x}{2}C_u\left[e^{i(\omega-\omega_L)t}+e^{-i(\omega+\omega_L)t}\right]
\end{eqnarray}
with $\omega_x = \mu B_{1,0}/\hbar$. Now, on \emph{resonance} ($\omega = \omega_L$) the above equations reduce to a set of coupled harmonic oscillators in $C_u$ and $C_d$. (We take the time average of the rapid oscillation $\left<e^{-i2\omega_Lt}\right> = 0$.) In other words, during the interval for which a resonant transverse field $B_1$ is applied, the spinor undergoes \emph{Rabi oscillations} between the two static eigenstates. By properly timing the duration of the $B_1$ ``pulse'', it is possible to invert the spin state (the so-called ``$180^\circ$ pulse''; i.e. $\ket{+} \rightarrow \ket{-}$ and vice versa), or even drive the spinor to a superposition state whose expectation lies in the transverse plane (a ``$90^\circ$ pulse''). In principle, such a vector remains in the superposition state indefinitely.

\subsection{Spin-lattice relaxation, $T_1$}

In the actual experiment, however, the samples are free to exchange energy with the environment (the ``lattice''). Therefore, the superposition state is energetically permitted to undergo wavefunction collapse into one of the eigenstates, thereby losing the component undergoing precession. The timescale of the return to the bias field axis is called the spin-lattice relaxation time $T_1$.

We have chosen to measure $T_1$ by using a $90^\circ-\tau-90^\circ$ pulse sequence. In this experiment, we first drive the spins into the transverse plane using a $90^\circ$ pulse. We then wait a period $\tau$, during which some of the precessing spins will have collapsed into the eigenstates. By applying a second $90^\circ$ pulse, we then bring the collapsed spins back into precession. At the same time, the second pulse sends the initially uncollapsed spins towards the bias field axis. By considering the size of the secondary FID, we can deduce the timescale of the spin-lattice relaxation.

\subsection{Spin-spin relaxation, $T_2$}

Even when the spin ensemble does not interact with the surroundings, the internal interaction among the spins can result in \emph{decoherence} of the precessive motion. As we have remarked earlier, the macroscopic Hamiltonian due to the static $B_0$-field should in principle result in perpetual precession. However, the actual Hamiltonian experienced by the spins includes microscopic effects such as the $\mu/r^3$ spin-spin interactions among the nuclei. Furthermore, the bias field inhomogeneity and diffusion of individual spins through the volume all contribute as to yield a complicated microscopic Hamiltonian that is inconsistent with the coherent, precessive motion of the macroscopic Hamiltonian.

Thus there is potential for loss of transverse magnetization without collapse into the eigenstates. This is known as the spin-spin relaxation $T_2$. Among the above effects, the most fundamental is the spin-spin interaction, since the other effects are not intrinsic to the system. Moreover, there exist experimental procedures for overcoming the latter effects, as we now show.

Consider a purely non-diffusive sample in an inhomogeneous bias field. We can partition the sample into sufficiently small regions for which the local field is well-approximated to be uniform. In such a partition, each region has a fixed frequency of precession. Note that even in the case of ideal, non-interacting spins, the overall transverse magnetization will then be seen to decay, because the spins of different regions will evolve out of phase. In 1950, E. Hahn showed that a $180^\circ$ pulse to such a system will precisely reverse the relative phase accumulation in the different regions, so that the in-phase dynamics is recovered (the ``spin echo'') at some time following the $180^\circ$ pulse.\cite{hahn1950} (See Fig. \ref{fig:apparatus}.)

We measure $T_2$ by the Carr-Purcell (CP) pulse sequence, consisting of the following pulses: $90^\circ-\tau-180^\circ-2\tau-180^\circ-2\tau-...$, where $\tau$ is a fixed waiting period. The ensemble responds to the CP sequence with a spin-echo halfway between each $180^\circ$ pulses. By choosing a small $\tau$, we can even diminish the effect of diffusion in the sample, since there will be little time for the spins to move in between pulses. By considering the decay of the echo heights, we then calculate the loss of transverse magnetization in a way independent of the bias field inhomogeneity and diffusion effects.

\section{Experimental Setup}

The experimental apparatus consisted of a $B_0 = 1770$ gauss permanent magnet, RF drive and detection circuit, and a PC-based oscilloscope-capture card as shown in Figure \ref{fig:apparatus}. 
\begin{figure}[htb]
\includegraphics[width=8cm]{apparatus.png}
\caption{Schematic of the experimental apparatus. The scope trace shows the spin echo that arises due to a $180^\circ$ pulse following the decay of the initial FID. Figure from \cite{nmrlg_old}\label{fig:apparatus}}
\end{figure}

\begin{figure*}[htb!]
\includegraphics[width=18cm]{100_glycerine}
\caption{The $90^\circ-\tau-90^\circ$ pulse method applied to a sample of 100\% glycerine. (Left) A single FID fit against $V(t) = \alpha e^{-\gamma t}\sin(\omega t + \phi) + \delta$. The decaying envelope is plotted and the extrapolated amplitude is placed with its corresponding error bar. (Right) FIDs of various $\tau$ are superimposed, and an exponential envelope is fitted.  From this regression, it is found that $T_1=25.0\pm0.6$ ms.
\label{fig:90_90}}
\end{figure*} 

The function generator provided a continuous sinusoid, yielding a RF drive of approximately $25$V(pp) at the probe. We generated pulses of desired duration by gating this signal with a switch, using logic from a digital pulse programmer. A major shortcoming of the apparatus was the limited precision with which we could time the pulses. In particular, we had a timing resolution of $1 \mu$s, which was not sufficient for the generation of $180^\circ$ pulses (and by implication also the $90^\circ$ pulses). When a $180^\circ$ pulse is applied to the thermal sample, we do not expect any FID. However, during the experiment, we found that it was impossible to adjust the parameters as to give a $180^\circ$ pulse that did not produce some FID. 

The gated RF signal was amplified and sent to the probe circuit, consisting of a solenoid wrapped around a vial containing the NMR sample. During the readout step of the experiment, the same solenoid was used to detect the precessing spins by magnetic induction.% The detected signal was amplified and filtered by a $7.5$MHz bandpass.

The multiplier and the LPF allowed us to quantitatively estimate the resonant frequency of the samples. In general, the applied frequency $\omega$ and the spin precession rate $\omega_L$ will not be identical. Recall that the effect of a multiplier is to produce a superposition of two sinusoids at the difference and sum frequencies. The LPF preserves only the difference frequency. Hence, the apparatus offered an observable benchmark for being ``on-resonance'': at resonance, we observed a severely band-limited signal on the oscilloscope. We were also able to obtain an error in our estimate of $\omega_L$ by examining the frequency content of the final signal via a Fourier transform.

For the remainder of the experiment, the multiplier/LPF chain was a slight detriment, because we were forced to purposely move off of resonance (order of kHz), in order to observe reasonable FIDs. Since our pulse durations were around $50\mu$s, this slightly invalidates our assumption of perfect Rabi oscillations over the pulse duration, i.e. that $e^{i(\omega-2\omega_0)t} = 1$ over the interval. 

Finally, a major limitation in our experiment was the rather slow sampling rate ($T_{sample} = 25\mu$s for the $T_2$ measurements) of the oscilloscope acquisition. In some of our runs, we have noticed significant discrepancies on the actual oscilloscope trace, and the PC-acquired data. This is troublesome for our estimates for the heights of peaks, since the actual maximum of a peak may have been aliased in the sampling process. We also note that the acquisition software converted the signal into arbitrary units; and that it introduced quantization errors of $0.6\times10^{-3}$ in the arbitrary units.

\section{Results and discussion}

\subsection{Determination of resonant frequencies and the gyromagnetic ratio}

We have performed ten measurements of the resonant frequency by the frequency differencing technique previously described, and found $\omega/2\pi = (7.522\pm0.049)$ MHz. Using the relation $\omega = \gamma B_0$ at resonance, and measurements of $B_0$, we have calculated the gyromagnetic ratio in terms of the g-factor. We obtained $g = 5.58\pm0.03$.

%\begin{table}[h]
%\caption{\label{tab:gfactor}The proton g-factor is computed from measurements of the
%Larmor frequency on a sample of 100\% glycerine.}
%\begin{ruledtabular}
%\begin{tabular}{cccc}
%Trial&$g$&$\omega/2\pi$ (MHz)&$B_0$ (kGauss)\\
%\hline
%1&$5.594\pm0.033$&$7.521\pm0.049$&$1.764\pm0.01$\\
%2&$5.569\pm0.033$&$7.522\pm0.049$&$1.772\pm0.01$\\
%3&$5.645\pm0.033$&$7.522\pm0.049$&$1.748\pm0.01$\\
%4&$5.575\pm0.033$&$7.522\pm0.049$&$1.770\pm0.01$\\
%5&$5.578\pm0.033$&$7.522\pm0.049$&$1.769\pm0.01$\\
%6&$5.571\pm0.033$&$7.522\pm0.050$&$1.771\pm0.01$\\
%7&$5.565\pm0.032$&$7.522\pm0.037$&$1.773\pm0.01$\\
%8&$5.575\pm0.033$&$7.522\pm0.049$&$1.770\pm0.01$\\
%9&$5.562\pm0.033$&$7.522\pm0.049$&$1.774\pm0.01$\\
%10&$5.553\pm0.032$&$7.522\pm0.049$&$1.777\pm0.01$\\
%\end{tabular}
%\end{ruledtabular}
%\end{table}

\subsection{Relaxation times vs. viscosity}

We have performed $T_1$ and $T_2$ measurement for various glycerine-water mixtures, using the $90^\circ-\tau-90^\circ$ and Carr-Purcell pulse sequences respectively. The degree of mixing (i.e. glycerine percent weight) was converted into a measure of viscosity via \cite{water-glycerine}. Figure \ref{fig:90_90} shows the typical growth of the secondary FID amplitude as a function of $\tau$ in the $T_1$ measurement. The results from various $\tau$ have been superimposed in order to illustrate the exponential characteristic of the spin-lattice relaxation.% By observing a part of the trace that contains no signal, we were able to estimate the noise level. The standard-deviation of the noise sample was taken to be the error associated with each sampled point.

To each of the secondary FIDs, we were able to fit a decaying sinusoid. We then extrapolated the exponential envelope to the beginning of the FID. The initial amplitude was taken to be the ``height'' of the secondary FID. As shown in Figure \ref{fig:90_90}, the exponential model was an excellent characterization of the relaxation process. This analysis was performed for each of the available mixtures, yielding $T_1$ values as function of viscosity.  

For the spin-spin relaxation measurements, we have used a much simpler numerical procedure. This was necessary due to several factors: (1) the double-sided exponential envelope proved to be a more difficult function to fit numerically; and (2) the spin echoes from the Carr-Purcell sequences were embedded in long data vectors that were difficult to parse reliably. With this in mind, Figure \ref{fig:t2_analysis} illustrates the procedure we have used to deduce $T_2$ from the Carr-Purcell response.

\begin{figure}[htb]
\includegraphics[width=8cm]{T2_analysis.png}
\caption{Illustration of the Carr-Purcell analysis on $98\%$ glycerine. The basic periodicity of the signal is deduced by considering the first two FIDs (top). Based on the computed period, the spin echoes are identified (middle). Finally, an exponential regression of the echo peaks is performed. Here, we obtain $T_2 = 20.0\pm2.2$ ms, and $\chi^2_{\nu-1} = 1.23$. \label{fig:t2_analysis}}
\end{figure}

We began by considering two adjacent spin echoes (top panel), from which the basic period was deduced. Using this spacing, we were able to grab each echo (middle). For each, we then computed the value of the maximum height (bottom). Note that this ``peak sampling'' technique for $T_2$ is more susceptible to aliasing issues than the regression procedure for $T_1$, since the latter method is sensitive to the internal phase of the oscillatory signal. We have chosen to employ an estimate for the error in the height determination (for $T_2$) that would compensate for the additional susceptibility. Namely, we consider $N$ neighboring points about the maximum, and use their standard deviation as the error. Note that if the sampled peak were not a true maximum in the original signal, we expect larger standard deviation in the set of neighborhood points, since the slope will be nonzero.
\begin{figure}[htb]
\includegraphics[width=8cm]{visc_results.png}
\caption{Relaxation times $T_1$ and $T_2$ as a function of sample viscosity. Bloembergen's original data for $T_1$ is also plotted for comparison. We have verified Bloembergen's observation of the inverse relationship between $T_1$ and viscosity. \label{fig:visc_results}}
\end{figure}

These procedures yielded $T_1$ and $T_2$ values as presented in Fig \ref{fig:visc_results}. For comparison, Bloembergen's original study of the viscosity dependence of $T_1$ is also plotted. It is evident that our measurements yield a slope parallel to Bloembergen's results. Hence, we confirmed Bloembergen's inverse-law observation between the relaxation times vs. viscosity. As we have predicted, additional microscopic interactions (a more viscous sample) increases the relaxation rates.

\subsection{Relaxation times vs. paramagnetic ion doping}

The above $T_1$ and $T_2$ fitting procedures were repeated on a sequence of samples that were doped with various $Fe^{3+}$ concentrations. The addition of these ions are expected to increase relaxation rates, since they contribute to microscopic spin-spin interactions within the sample. The results are plotted in Fig. \ref{fig:para_results}.
\begin{figure}[htb]
\includegraphics[width=8cm]{para_results_clean.png}
\caption{Relaxation times $T_1$ and $T_2$ as a function of impurities. Bloembergen's original data for $T_2$ is also plotted for comparison. Note the threshold effect near $10^{17}$ ions/cc. \label{fig:para_results}}
\end{figure}

Unfortunately, only limited comparisons to Bloembergen's data can be made, because the available range of concentrations did not overlap well with Bloembergen's samples. However, in the overlapping regions, we have found plausible agreement with the original study. In addition, below $10^{17}$ ions/cc levels, we have observed an interesting saturation behavior of the relaxation times. There are many possible interpretations of this phenomenon. The plateau may reflect a threshold effect in the impurity concentration. It may also be the case that some imperfection within our apparatus placed a ceiling in the relaxation times at the $1$s range. Or, in the worst case, it may signify flawed numerical procedures or defective samples. In general, however, the basic claim that increased microscopic complexity leads to faster relaxation rates is supported by our measurements.

\section{Error analysis}

We have resorted to the ``peak-sampling'' technique for measuring heights of spin echoes due to its simplicity. Ideally, the pulses should be characterized by a full regression, as was performed for the $T_1$ analysis. The main advantage is that the latter procedure is sensitive to the internal phase of the oscillatory signal, and is therefore less susceptible to aliasing issues in the sampling process. In particular, we have performed both techniques on $T_1$ data, and have found up to $100$ms discrepancies in the resulting relaxation times. 

We remarked previously that an estimate of the peak error (due to aliasing) may be obtained by considering the neighborhood of the maximal point. Such errors were typically on the order of $0.005$ (in the arbitrary units of the sampling process). This can be compared to the intrinsic digitization error of $0.0006$ of our equipment. Hence, if the proper fittings are performed in the spin echoes, it is in principle possible to obtain a factor of ten reduction in the echo height errors.

More fundamentally, we have observed that the envelope of each FID and spin echoes are not perfect exponentials. Some of this deviation may be observed in the FID of Fig. \ref{fig:90_90} where the tail shows discrepancies in the expected amplitude, as well as the oscillation frequency. Thus, we suspect that our fitting functions $V(t) = \alpha e^{-\gamma t}\sin(\omega t + \phi) + \delta$ may be theoretically incomplete. 

We have already discussed several instrumental shortcomings. In addition, we also believe that some of the paramagnetic ion samples may have been improperly prepared. For instance, precipitation was observed in many of the samples, which is clearly inappropriate for our experiment.

\section{Conclusions}

We have investigated relaxation phenomena in a sample of nuclear spins. The spin-lattice relaxation $T_1$ and the spin-spin relaxation $T_2$ times were measured as a function of sample viscosity and impurity concentrations. For the former, we have verified Bloembergen's observation that the relaxation times are inversely related to the sample viscosity. The study of impurity effects remains inconclusive. We require more samples in the range of Bloembergen's original study for a better assessment. In general, our experiment corroborates the intuition that additional microscopic interactions (increased viscosity, additional impurities) increases relaxation rates.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Place all of the references you used to write this paper in a file
% with the same name as following the \bibliography command
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\bibliography{thebibliography}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{acknowledgments} As usual, THK gratefully acknowledges Connor McEntee for his partnership in the experiment.
\end{acknowledgments}

\end{document}
